Author's response to reviews

Title: Aqueous extract of dioscorea opposita thunb. normalizes the hypertension in 2K1C hypertensive rats

Authors:

  Nurmuhammat Amat (nur818@hotmail.com)
  Raziya Amat (272855234@qq.com)
  Sajida Abdureyim (sajiedaa@hotmail.com)
  Parida Hoxur (Pld245@sina.com)
  Zulpiya Osman (zulpiya-525@163.com)
  Dolkun Mamut (dolkunmamut@hotmail.com)
  Anake Kijjoa (ankijjoa@icbas.up.pt)

Version: 2
Date: 7 November 2013

Author's response to reviews: see over
The Editor

*BMC Complementary and Alternative Medicine*

I hereby wish to submit the manuscript entitled “*Anti-hypertensive effects of the aqueous extracts of Dioscorea opposita Thunb. (yam) in Two-Kidney, One-Clip Renal Hypertension Rat model*” for a publication in the *BMC Complementary and Alternative Medicine*.

I would like to declare that the manuscript has not been previously published, is not currently submitted for review to any other journal, and will not be submitted elsewhere before a decision is made by this journal.

There are no exists conflict of interest in our paper. I would like also to declare that this manuscript has never been submitted for publication elsewhere and that all the listed authors have read and approved it. I also hope that the manuscript is prepared accordingly.

If you any queries about this manuscript, please fell free to contact me.

Sincerely Yours,

Dr Nurmuhammat.Amat

*We revised the article according to reviewers commenments.*

1. Rewrite the figure legends.
2. Exact number of animals used for each group should be included in each figure legends.
3. Source of captopril (drug) should be provided in Material and Method part.
4. In the result part, we corrected the “DOT significantly regulated the antioxidant status of 2K1C hypertensive rats” described in figure.
5. We add the captopril to material section.